# UC Riverside UC Riverside Previously Published Works

# Title

Earth System Model Analysis of How Astronomical Forcing Is Imprinted Onto the Marine Geological Record: The Role of the Inorganic (Carbonate) Carbon Cycle and Feedbacks

**Permalink** https://escholarship.org/uc/item/5cw6j3h5

**Journal** Paleoceanography and Paleoclimatology, 36(10)

**ISSN** 2572-4517

# **Authors**

Vervoort, Pam Turner, Sandra Kirtland Rochholz, Fiona <u>et al.</u>

Publication Date 2021-10-01

# DOI

10.1029/2020pa004090

Peer reviewed



**f** 

# Paleoceanography and Paleoclimatology®

# **RESEARCH ARTICLE**

10.1029/2020PA004090

#### **Key Points:**

- Time-varying astronomical forcing simulated on a multimillion-year timescale using an Earth system model of intermediate complexity
- Strong 100 kyr power present in temperature and wt% CaCO<sub>3</sub> is absent from  $\delta^{13}$ C cycles in the inorganic carbon reservoir
- Additional organic carbon feedbacks and a reduced weathering feedback likely played a role in early Cenozoic climate-carbon cycle dynamics

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

P. Vervoort, pam.vervoort@ucr.edu

#### Citation:

Vervoort, P., Kirtland Turner, S., Rochholz, F., & Ridgwell, A. (2021). Earth system model analysis of how astronomical forcing is imprinted onto the marine geological record: The role of the inorganic (carbonate) carbon cycle and feedbacks. *Paleoceanography and Paleoclimatology*, *36*, e2020PA004090. https://doi. org/10.1029/2020PA004090

Received 18 AUG 2020 Accepted 11 SEP 2021

#### © 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Earth System Model Analysis of How Astronomical Forcing Is Imprinted Onto the Marine Geological Record: The Role of the Inorganic (Carbonate) Carbon Cycle and Feedbacks

## Pam Vervoort<sup>1</sup> , Sandra Kirtland Turner<sup>1</sup>, Fiona Rochholz<sup>2,3</sup>, and Andy Ridgwell<sup>1</sup>

<sup>1</sup>Department of Earth and Planetary Science, University of California, Riverside, CA, USA, <sup>2</sup>MARUM Center for Marine and Environmental Sciences, University of Bremen, Bremen, Germany, <sup>3</sup>Now at Department of Geography, Research Group for Earth Observation, Heidelberg University of Education, Heidelberg, Germany

**Abstract** Astronomical cycles are strongly expressed in marine geological records, providing important insights into Earth system dynamics and an invaluable means of constructing age models. However, how various astronomical periods are filtered by the Earth system and the mechanisms by which carbon reservoirs and climate components respond, particularly in absence of dynamic ice sheets, is unclear. Using an Earth system model that includes feedbacks between climate, ocean circulation, and inorganic (carbonate) carbon cycling relevant to geological timescales, we systematically explore the impact of astronomically modulated insolation forcing and its expression in model variables most comparable to key paleoceanographic proxies (temperature, the  $\delta^{13}$ C of inorganic carbon, and sedimentary carbonate content). Temperature predominately responds to short and long eccentricity and is little influenced by the modeled carbon cycle feedbacks. In contrast, the cycling of nutrients and carbon in the ocean generates significant precession power in atmospheric CO<sub>2</sub>, benthic ocean  $\delta^{13}$ C, and sedimentary wt% CaCO<sub>3</sub>, while inclusion of marine sedimentary and weathering processes shifts power to the long eccentricity period. Our simulations produce reduced pCO<sub>2</sub> and dissolved inorganic carbon (DIC)  $\delta^{13}$ C at long eccentricity maxima and, contrary to early Cenozoic marine records, CaCO, preservation in the model is enhanced during eccentricity-modulated warmth. Additionally, the magnitude of  $\delta^{13}$ C variability simulated in our model underestimates marine proxy records. These model-data discrepancies hint at the possibility that the Paleogene silicate weathering feedback was weaker than modeled here and that additional organic carbon cycle feedbacks are necessary to explain the full response of the Earth system to astronomical forcing.

**Plain Language Summary** The Earth's tilt and orbit around the Sun change periodically, influencing the amount and distribution of solar energy that reaches the surface over time. We use a numerical model to simulate how changes in solar radiation impact the surface climate, ocean circulation, and the carbon cycle. Our model output shows that global mean surface temperatures are mainly controlled by eccentricity (a measure for how elliptical Earth's orbit is), changing every 100,000 and 400,000 years. Changes in the carbon cycle are mainly controlled by precession (a measure for the direction in which the Earth is tilted relative to the Sun), changing approximately every 20,000 years. However, precession cycles are not well preserved in geological records because, (a) the large total mass of carbon in the atmosphere and ocean means that small fluxes over 20,000 years are insufficient to cause drastic changes, and (b) mixing of marine sediments distorts the signal. We also find that modeled variations in the climate-carbon cycle are smaller than those observed in geological records. This suggests that other processes not modeled here, such as burial and release of organic carbon, are important to explain the full astronomical climate-carbon cycle variability.

## 1. Introduction

High-resolution paleoclimate records across the Cenozoic Era show that global climate and carbon cycling are closely connected on astronomical timescales on the order of  $10^4$ – $10^6$  years (e.g., Barnet et al., 2019; Mix et al., 1995; Pälike et al., 2006; Peterson et al., 2014; Sigman & Boyle, 2000; Turner et al., 2014; Wang et al., 2010; Westerhold et al., 2018; Zachos et al., 2001). In particular, during the early, ice-free Cenozoic,

benthic foraminiferal oxygen and carbon isotopes ( $\delta^{18}$ O and  $\delta^{13}$ C) show a strongly coherent astronomical influence (Cramer et al., 2003; Littler et al., 2014; Turner, 2014; Westerhold et al., 2011; Zachos et al., 2010), suggesting tightly coupled carbon cycle-climate dynamics. Multiple records reveal periodic, astronomically paced negative  $\delta^{13}$ C excursions with a magnitude of 0.3%–1.0% coinciding with deep water warming of 2°C–4°C as well as reductions in sedimentary calcium carbonate content (Lourens et al., 2005; Sexton et al., 2011; Zachos et al., 2010). Explanations for an astronomical trigger for these events, known as hyperthermals, invoke dynamics of a reduced carbon reservoir such as methane hydrates (Dickens, 2003; Lunt et al., 2011), permafrost (DeConto et al., 2012), wetlands (Kurtz et al., 2003; Zachos et al., 2010), or marine dissolved organic matter (Sexton et al., 2011), potentially in combination with thresholds in the climate system (Lunt et al., 2011).

Variations in global climate and the carbon cycle during the early, ice-free Cenozoic, including "hyperthermals," are predominately paced by short and long eccentricity cycles (Barnet et al., 2019; Lauretano et al., 2015; Littler et al., 2014; Turner, 2014; Westerhold et al., 2011, 2018; Zachos et al., 2001). While eccentricity is the only astronomical parameter that alters Earth's annual global mean solar energy insolation, the total variability is small (~0.5 Wm<sup>-2</sup>) (Laskar et al., 2004). Local seasonal insolation forcing related to precession is much stronger (Berger, 1978), and therefore one hypothesis is that carbon-climate variability is driven by precession forcing but modulated by eccentricity cycles (e.g., Maslin & Ridgwell, 2005; Zeebe et al., 2017). Additionally, the large mass of dissolved inorganic carbon in the ocean should act as a lowpass filter on carbon cycle variability, producing muted precession signals relative to eccentricity (Cramer et al., 2003; Pälike et al., 2006).

Numerical models are key tools for testing hypotheses about the origin of astronomically forced climate-carbon cycles. Previous studies have utilized atmosphere-ocean global climate models (GCMs) to reveal the mechanisms by which the astronomical configuration exerts a significant control on surface dynamics. For instance, the GENESIS GCM (Thompson & Pollard, 1997) was used to evaluate Eocene (paleo)oceanographic differences between two precession configurations (Sloan and Huber, 2001). That same model (with a slab-ocean) was used to reconstruct the extent of Eocene permafrost under various astronomical configurations, providing evidence for astronomically triggered carbon release at high latitudes (DeConto et al., 2012). Lunt et al. (2011) used the fully coupled HadCM3L GCM to show differences in ocean circulation under certain astronomical configuration, suggesting that variations in intermediate water mass temperature might drive the periodic destabilization of methane hydrates.

Because GCMs are computationally expensive, simulations usually apply fixed astronomical parameters, as in the studies described above, with the consequence that results cannot be compared directly to spectral analyses of proxy time series data. Exceptions include simulation of the complete deglacial transition with transient astronomical forcing (Liu et al., 2009), although this time interval is too short to capture complete astronomical cycles. Other studies have employed simpler box models to simulate the transient effects of astronomical forcing on the carbon cycle, typically using artificial forcing functions in the form of Eccentricity-Tilt-Precession (Laurin et al., 2015; Ma et al., 2011), or seasonal insolation curves (Pälike et al., 2006; Zeebe et al., 2017). While such models can simulate the transfer of spectral power amongst reservoirs and capture potential spectral shifts, the construction of the forcing functions requires basic presumptions about the mechanisms that drive the climate response.

Earth system Models of Intermediate Complexity (EMICs) bridge the gap between the most complex GCMs and box models that lack a dynamic ocean. Their computational efficiency allows generation of large ensembles of experiments and/or long integration time. EMIC studies of astronomical forcing, including those with the LOVECLIM (Goosse et al., 2010), CLIMBER-2 (Ganopolski et al., 2001; Petoukhov et al., 2000), and UVic (Weaver et al., 2001) models, have focused on the Quaternary (Bounceur et al., 2015; Brovkin et al., 2002; Calov et al., 2005; Heinemann et al., 2014; Konijnendijk et al., 2011; Menviel et al., 2011; Xiao et al., 2013) or the Pliocene (Willeit et al., 2013). While these studies demonstrate the sensitivity of the climate and carbon cycle to insolation forcing, a modeling framework that systematically evaluates the imprint of astronomical forcing in various components of the carbon cycle is missing. Moreover, even these model simulations, with time-varying orbital parameters, have been too short to evaluate the spectral power in model output on astronomical timescales (>10<sup>5</sup> years). There remains a need to evaluate how astronomical

forcing translates into the variables ultimately recorded by proxies (such as  $\delta^{18}O$ ,  $\delta^{13}C$ , and CaCO<sub>3</sub> content) on these timescales.

Here, we take a first step in this direction, driving the "cGENIE" EMIC (Ridgwell et al., 2007) with time-varying astronomical forcing over multimillion-year simulations. We develop and present a framework to assess the transient response of the Earth's climate-carbon system to astronomical forcing under ice-free greenhouse conditions, by separately quantifying the impact of insolation forcing on climate variability due to feedbacks in ocean circulation, marine productivity, CaCO<sub>3</sub> compensation, and terrestrial weathering. The simulations presented here focus on feedbacks involving the marine inorganic (carbonate) carbon cycle. A follow-up companion paper will separately interrogate the role of temperature-dependent feedbacks in the marine organic matter cycle. Our experimental framework hence provides a systematic model evaluation of the propagation of spectral signals through the Earth system to facilitate comparison to proxy datasets.

## 2. Methods

The cGENIE model combines a dynamic 3D ocean model with a simplified 2D energy-moisture-balance atmosphere model (Edwards & Marsh, 2005). cGENIE includes a detailed representation of ocean biogeochemical cycles (Ridgwell et al., 2007) plus the long-term (inorganic) carbon cycle represented through input and removal of dissolved carbon species via terrestrial rock weathering and calcium carbonate burial in marine sediments (Colbourn et al., 2013; Ridgwell & Hargreaves, 2007). This configuration of cGENIE has previously been employed on timescales up to 1 Myr to assess past climate events (e.g., Gutjahr et al., 2017) as well as the long-term future carbon cycle response to fossil fuel carbon release (e.g., Lord et al., 2016; Winkelmann et al., 2015). Our methodological advance is to include spatiotemporal insolation forcing based on astronomical parameters using equations from Berger (1978). Parameters for eccentricity, obliquity, and precession (longitude of perihelion) are updated every 1,000 years and the daily mean insolation at each latitude is recalculated accordingly. We select parameters from the La2004 astronomical solution for an early Cenozoic time slice between 57 and 53 Ma (Laskar et al., 2004) and assume a solar constant of 1361.7 Wm<sup>-2</sup>, consistent with this interval.

## 2.1. Continental Configuration

We employ a pair of deliberately idealized continental configurations (Figure 1) in order to distinguish climatic effects related to the carbon cycle from any astronomically forced effects that relate to the specific distribution of landmasses (e.g., Short et al., 1991). This provides a more generalized understanding of the ice-free Earth system response to astronomical forcing than would have been possible, if we had chosen a specific continental configuration. However, it should be noted that it is not possible to understand all the ways in which paleogeography impacts atmosphere and ocean circulation and how these patterns may interact with astronomical forcing, using only a pair of end-member configurations.

In the first configuration (symmetric configuration [SYMM]), a single continent spans pole to pole, creating a hemispherically symmetric landmass distribution (Vervoort et al., 2019). In the second, asymmetric configuration (ASYM), a single super-continent covers the northern polar region. While no Cenozoic continental plate arrangement closely reflects either end-member, we focus primarily on results using the ASYM setup because of the relative bias in the Northern Hemisphere landmass distribution throughout the Cenozoic. Comparable results for all SYMM experiments are included in the SI.

Because our model configuration lacks a dynamical atmosphere, a simplified zonally averaged wind-stress profile is generated and applied to the ocean surface with matching wind products used in the atmospheric advection of heat and moisture and in calculating air-sea gas exchange. Similarly, a meridional albedo profile is prescribed independent of the continental configuration but as a function of latitude only with the highest albedo values in the high latitudes, where the angle of incoming solar radiation enhances reflectivity. While differences in continental configuration should correspond to differences in albedo, our cGENIE experiments lack a land surface model to provide a more realistic pattern. Both SYMM and ASYM continental configurations consist of  $18 \times 18$  equal-area grids and 16 ocean depth levels of exponentially increasing thickness with depth. We truncate the vertical grid at 14 levels, giving a maximum depth of ~3,575 m to approximate modern ocean volume and adopt a flat-bottom bathymetry. We generate and apply





**Figure 1.** cGENIE configurations. (a and b) The continental landmass distribution (gray), prescribed wind stress fields (vectors), albedo profiles, and simulated sea surface temperature (SST, color coded) for (a) the asymmetric configuration (ASYM) and (b) symmetric configuration (SYMM). (c and d) The zonal mean ocean overturning strength indicated by contour lines, dashed for a counter-clockwise circulation, solid lines for clockwise movement. Carbonate ion  $(CO_3^{2-})$  concentrations for (c) ASYM and (d) SYMM are color coded. The bottom panels include normalized power spectra for (e) the annual global mean insolation and (f) the annual zonal mean insolation.

a random spatial distribution of ocean sediment depths consistent with the modern hypsometric curve that is used in the pressure-dependent calculation of calcium carbonate preservation. Boundary conditions were generated using the "muffingen" software suite v0.9.25 (https://doi.org/10.5281/zenodo.5130677) and are shown in Figure 1 and S1.

#### 2.2. cGENIE Spin-Up Stages

The model is equilibrated in a two-step spin-up process for each of the SYMM and ASYM configurations. An initial 20,000-year spin-up equilibrates ocean dynamics and ocean biogeochemical cycling with atmospheric CO<sub>2</sub> concentrations fixed at 834 ppm (Ridgwell & Schmidt, 2010), emulating an early Cenozoic "warm



Table 1         Experimental Design				
	Climate-CO <sub>2</sub> feedback	Marine export feedback	CaCO <sub>3</sub> compensation	Terrestrial weathering
Exp.0	-	-	-	Tracking
Exp.1	1	-	-	Tracking
Exp.2	1	1	-	Tracking
Exp.3	1	1	1	Fixed
Exp.4	1	1	1	Responsive

*Note.* For Exp.0–4, shown which carbon-climate feedbacks are excluded (-) and included ( $\checkmark$ ). All experiments are performed under SYMM (hemispherically symmetric continent) and ASYM (supercontinent at the Northern Hemisphere) configurations. Note that marine surface productivity is present, but unresponsive to astronomical forcing in Exp.0 and 1 while surface productivity varies with astronomical forcing in Exp.2–4. For terrestrial weathering, 'tracking' indicates that weathering fluxes are exactly equal to (variable) CaCO<sub>3</sub> burial fluxes, 'fixed' means weathering fluxes are fixed and decoupled from CaCO<sub>3</sub> burial, "responsive" indicates weathering rates vary in response to changes in land temperature.

climate" and yielding sufficiently high temperatures to prevent both perennial and seasonal sea ice formation at the high latitudes. Values for the astronomical parameters are fixed during spin-up at their average values across the modeled interval (53–57 Ma). The initial values of mean ocean alkalinity (ALK), dissolved inorganic carbon (DIC), nutrient (phosphate) concentrations, and major ion concentrations of  $Ca^{2+}$ ,  $Mg^{2+}$ , and  $SO_4^{-2-}$  of the model follows previous Paleocene cGENIE configurations (e.g., Ridgwell & Schmidt, 2010).

In the second stage spin-up, also with fixed astronomical parameters, the long-term carbon cycle is enabled by allowing loss of solutes from the ocean through marine  $CaCO_3$  burial and balancing this with a terrestrial weathering flux. Atmospheric  $CO_2$  is stabilized close to 834 ppm by adjusting the balance between volcanic outgassing and terrestrial silicate weathering (dependent on land temperature as described in Colbourn et al., 2013 and Lord et al., 2016). The final steady states of ocean carbonate chemistry differ slightly between SYMM and ASYM, a consequence of the differences in continental distribution and hence large-scale pattern of ocean circulation (Figures 1c and 1d, Table S1).

#### 2.3. Experimental Design

Starting from these initial steady states, the effects of astronomical forcing on climate and the carbon cycle are systematically investigated, run-

ning cGENIE transiently for 4 million years to allow the evaluation of astronomical cycles with periods up to 500 kyr. We conduct a set of five simulations for both SYMM and ASYM in which the complexity of the carbon cycle is sequentially increased (Table 1).

In the simplest experiment (Exp.0), we disable all carbon cycle feedbacks and fix radiative forcing with respect to an atmospheric  $CO_2$  composition of 828 ppm for ASYM and 834 ppm for SYMM (Table S1). Variability in global climate is thus only a function of the temporal radiative balance and feedbacks relating to physical processes, such as meridional ocean and atmospheric heat transport, and ocean-atmosphere exchange of heat and moisture. Cycles produced in atmospheric  $CO_2$  have no influence on the modeled temperature variability.

In the subsequent experiment (Exp.1), we allow variations in atmospheric CO<sub>2</sub> to modulate surface temperature. The magnitude and spatial distribution of export production of both particulate organic carbon (POC) and CaCO<sub>3</sub> is fixed, and weathering fluxes are forced to exactly track CaCO<sub>3</sub> burial such that the ocean ALK and calcium ion inventories become invariant. Global climate now additionally responds to carbon cycle feedbacks related to the controls of ocean circulation and solubility on ocean CO, uptake. Resultant changes in ocean carbonate chemistry will impact the preservation of CaCO<sub>3</sub> (but does not cause any change in ocean ALK and calcium ion inventories) plus the distribution of DIC  $\delta^{13}$ C. With fixed export, preservation of CaCO<sub>3</sub> in surface sediments varies as a function of temperature, salinity, and (CO<sub>3</sub><sup>2-</sup>) of overlying seawater.  $\delta^{13}C$  of DIC is controlled by multiple fractionation processes. First, isotopic partitioning occurs during re-equilibration between CO2, HCO3-, and CO3- (Ridgwell et al., 2007; Zeebe & Wolf-Gladrow, 2001). A second temperature-dependent isotopic fractionation occurs during the air-sea gas exchange (Ridgwell et al., 2007; Zhang et al., 1995). We also model the fractionation associated with organic matter formation as a function of aqueous CO<sub>2</sub> concentration (Rau et al., 1996; Ridgwell, 2001) as well as temperature-dependent isotopic fractionation between (HCO<sub>3</sub><sup>-</sup>) and CaCO<sub>3</sub> (Mook, 1986). See Turner and Ridgwell (2016) for a detailed description of the representation of  $\delta^{13}$ C in cGENIE including evaluation of simulated spatial patterns.

In Exp.2, POC (with  $CaCO_3$  in a fixed ratio of 0.2 mol:mol) export is now allowed to vary in space and time as a function of light and nutrient (phosphate) availability following Ridgwell et al. (2007). Exported POC is remineralized through the water column according to a fixed exponential decay depth profile (Ridgwell et al., 2007). Again, all carbon, ALK, and calcium removed from the system through  $CaCO_3$  burial is automatically replenished with an equal flux from terrestrial weathering. This configuration provides ocean-atmosphere carbon cycle feedbacks but no geological carbon feedbacks.

We add the first geological carbon feedback in Exp.3 by decoupling terrestrial weathering (providing the alkaline building blocks for  $CaCO_3$  production) from marine  $CaCO_3$  burial. Weathering rates are fixed at the values obtained from the end of the open system spin-up, while  $CaCO_3$  preservation varies based on  $CaCO_3$  export productivity and the saturation state of overlying water. The ocean ALK (and  $Ca^{2+}$ ) inventory can now vary and impact  $pCO_2$  via 'carbonate compensation'.

Finally, Exp.4 completes the representation of the long-term (inorganic) carbon cycle by including a terrestrial weathering feedback (see Lord et al., 2016 for an analysis of the modes of response and characteristic timescales of the complete system). Carbonate weathering varies linearly with mean global air temperature over land whereas silicate weathering responds exponentially (Colbourn et al., 2013).

To elucidate the role of each set of carbon cycle feedbacks in the Earth system response to astronomical forcing, we investigate the difference between each set of experiments, for example, output of Exp.3 is subtracted from output of Exp.4 to isolate the impact of the astronomically forced weathering feedback on the Earth system. Hence while Figures 3d-3f presents the response in  $pCO_2$ , benthic DIC  $\delta^{13}C$ , and wt% CaCO<sub>3</sub> that results from the astronomically forced changes in ocean circulation and the solubility feedback (Exp.1), subsequent Figures 4d-4f through 6d-6f) are shown in the form of anomaly plots for the difference from the previous experiment. Likewise, we plot the change in the power spectra ( $\Delta$ FFT) with respect to the previous experiment to demonstrate the relative weakening (negative power) or amplification (positive power) of astronomical cycles when adding feedbacks.

#### 2.4. Timeseries Analysis

Spectral analysis was performed on model output variables with a simple Fast Fourier Transform (FFT) to extract the dominant cycles (shorter than 500 kyr) and their average ranges. The average range is equal to twice the average amplitude across the 4 million year interval for each spectral peak determined by the FFT. Because model output is (white) noise-free in the absence of internal interannual variability in cGENIE, further analysis to determine the statistical significance of the astronomical peaks is not required. The phasing between the imposed short and long eccentricity forcing and cycles in model variables of interest are determined using a multi-taper coherence method (Table S2). Because model output is saved at 1000-year intervals over the 4 Myr long experiments, we are unable to diagnose any lags with a duration less than 1,000 years.

## 3. Results and Discussion

In the following sections we describe the main results of Exp. 0–4, separately assessing the role of specific carbon cycle feedbacks in the Earth system response to astronomical forcing in proxy relevant model outputs, including ocean temperature, atmospheric  $pCO_2$ , DIC  $\delta^{13}C$ , and sedimentary wt% CaCO<sub>3</sub>. We focus on mean annual output and evaluate signals in terms of the total variability generated, distribution of spectra power, the average range of individual cycles across the 4 million year interval, and phasing with respect to astronomical forcing and between model output. Results of the asymmetric (ASYM) world are presented in the main figures, while comparison with the equivalent SYMM simulations is provided in the final paragraph of each subsection. Corresponding graphical SYMM analyses are provided as Figures S2–S8. See Table S2 for a summary of cycle ranges and phasing calculated for each experiment. Comparison between ASYM and SYMM results allows us to evaluate how latitudinal heterogeneity in the landmass distribution impacts the global response to astronomical forcing. Figure 1 shows the power spectra of mean annual insolation power spectra is dominated by eccentricity, while for most latitudes, obliquity dominates mean annual power spectra.





**Figure 2.** Astronomical climate evolution in asymmetric configuration (ASYM) of Exp.0. (a) The eccentricity, obliquity, and the precession index ( $e \sin \omega$ ) used to calculate the daily mean insolation across 4 million years of cGENIE simulation. (b) The simulated change in the global annual mean temperature in the atmosphere (black), surface (light blue), and benthic ocean (dark blue) with their respective normalized power spectra in panels (c-e) calculated using the Fast Fourier Transform. Each value near a spectral peak indicates the associated period of the cycle in kyr. The power spectra of temperature in the (f) atmosphere, (g) surface, and (h) deep ocean are shown for each latitude.

#### 3.1. Global Climate Response

In Exp.0 (Figure 2), annual global mean air temperature responds to annual global mean insolation forcing but with no CO, feedback on climate. In the ASYM configuration, the maximum variability in atmospheric temperature in response to astronomical forcing is 1.7°C. Spectral power in the global mean annual air temperature is dominated by the 400 kyr (long eccentricity) and 125 and 95 kyr (short eccentricity) periods (Figure 2c). The distribution of spectral power in global mean annual air temperature is similar to the distribution of spectral power in mean annual temperature at individual latitudes (Figure 2f). Seasonal averaging removes power in Earth's mean annual temperature response at higher frequencies, and local mean annual temperature does not respond linearly to the local mean annual insolation (contrast Figure 1f with Figure 2f). Mean annual insolation at most latitudes is dominated by obliquity power (Figure 1f), and obliquity power also appears in global mean annual air temperatures. In ASYM simulations, small precession cycles in the global mean annual air temperature result from the different heat capacity of each hemisphere and are consistent with the dominance of precession power in mean annual air temperature over the modeled Northern Hemisphere (NH) landmass (also recognized by Short et al., 1991) (Figure 2f). The NH, where the continent is located, has a smaller heat capacity than the Southern Hemisphere (SH) and thus the increase in the global annual mean temperature is greater when the NH is tilted toward the Sun at perihelion, compared to when the SH is tilted toward the Sun at perihelion.

Changes in the surface air temperature are instantaneously transferred (with a lag less than the 1 kyr interval of model data saving) to the surface ocean. Although the maximum variability in global mean annual sea surface temperature (SST) is slightly smaller than global mean annual surface air temperature ( $1.3^{\circ}$ C vs.  $1.7^{\circ}$ C), the distribution of spectral power is nearly identical (Figures 2c and 2d). Astronomically forced temperature changes in the benthic ocean are slightly reduced compared to the surface ocean, with a maximum variability of  $1.1^{\circ}$ C and spectral power is dominated by eccentricity (Figure 2e) with respective average ranges of  $0.25^{\circ}$ C and  $0.30^{\circ}$ C for the short and long eccentricity cycles. The major difference between the SST and benthic ocean temperature power spectra is a loss of power at obliquity and precession frequencies. On a global mean, benthic ocean temperature lags eccentricity forcing by ~1 kyr, consistent with the timescale of ocean overturning.

Astronomically forced climate evolution in SYMM is comparable to ASYM in Exp.0 with the exception that the maximum variability in temperature is slightly reduced and precession cycles are absent (Figure S2). We attribute the loss of precession power to the similar heat capacity of both hemispheres in SYMM.

#### 3.2. Ocean Solubility Pump

Astronomical forcing induces spatiotemporal variations in ocean temperature (Figure 2), and alters the strength of ocean overturning circulation. By enabling a  $CO_2$ -climate feedback in Exp.1 (Figure 3), changes in atmospheric  $pCO_2$  resulting from the influence of solubility and ocean circulation on ocean carbon uptake can now modulate the direct temperature response to astronomical forcing (cf. Exp.0).

The power spectrum of modeled global ocean overturning is dominated by precession (Figure 3c). Ocean overturning is a seasonal process driven by maximum deep-water formation close after the coldest months at latitudes where seasonal insolation is dominated by precession power. Precession periodically facilitates relatively more intense overturning when the SH winter occurs at aphelion compared to perihelion. Enhanced deep-water formation and increased ventilation of the ocean interior physically transports respired



**Figure 3.** Exp.1, ocean circulation and  $CO_2$  solubility feedbacks in asymmetric configuration (ASYM). (a) Astronomical forcing parameters (eccentricity, obliquity (in degrees), and precession index (e sin  $\varpi$ )) and their Fast Fourier Transform (FFT) normalized to the highest individual power. (b) Change in benthic ocean temperature. (c) Change in the maximum Southern Hemisphere ocean overturning strength, defined by the maximum overturning strength reached in the SH across all ocean depth levels. (d) Change in atmospheric  $CO_2$ . (e) Change in benthic  $\delta^{13}C$  of dissolved inorganic carbon (DIC). (f) Change in wt% CaCO<sub>3</sub>. All variables are annual global mean values and are accompanied by their respective FFTs, normalized to the peak with maximum power.

carbon back to the surface ocean, and thereby slightly increases atmospheric  $pCO_2$ . However, the relatively small changes in the strength of ocean overturning (maximum variability of ~1 Sv, corresponding to a change in the maximum ocean ventilation age of ~100 years) are insufficient to drive major redistributions of carbon between the ocean and atmosphere. The average range of precession cycles in  $pCO_2$  is <3 ppm (Figure 3d).

The  $pCO_2$  power spectra is instead dominated by long eccentricity with a significant contribution of power at short eccentricity and obliquity. The total variability is ~32 ppm with average ranges of obliquity and short eccentricity cycles in  $pCO_2$  of ~5 ppm and ~6 ppm for long eccentricity cycles, resulting from feedbacks due to ocean solubility and carbon uptake. More  $CO_2$  is released from the surface ocean into the atmosphere at eccentricity maxima when global annual mean temperatures are highest (Figure 3d). However,  $CO_2$  uptake is not spatially or seasonally uniform and is most significant at cold months in the high latitudes, resulting in obliquity cycles with a range comparable to eccentricity cycles.

The global mean benthic DIC  $\delta^{13}$ C power spectrum shows maximum variability of 0.09‰. Obliquity and precession dominate the power spectrum with very little power at eccentricity (Figure 3e). The lack of eccentricity power in  $\delta^{13}$ C timeseries demonstrates that the local and seasonal changes in temperature are particularly significant controls on the modeled fractionation processes. Precession power is consistent with the impact of ocean overturning on the physical transport of low- $\delta^{13}$ C remineralized carbon from the deep to the surface ocean, generating cycles with an average range of ~0.02‰ that are anti-phased between the surface and deep ocean (Figure 3e). Obliquity power arises primarily due to changes in fractionation associated with export production and generates cycles with a comparable average range of ~0.02‰ also anti-phased with surface ocean DIC  $\delta^{13}$ C cycles. Reduced fractionation in the surface ocean leads to isotopically lighter surface DIC while relatively heavier carbon respires at depth which increases the benthic DIC  $\delta^{13}$ C. Although the magnitude and spatial patterns of export are fixed, fractionation in the formation of organic matter dependent, both of which drive changes in  $\delta^{13}$ C in response to obliquity-paced changes in temperature and air-sea gas exchange.

The power spectrum for global mean marine sedimentary wt% CaCO<sub>3</sub> closely resembles that of atmospheric CO<sub>2</sub>, but with slightly amplified obliquity and precession power (Figures 3d and 3f). Strong eccentricity power in wt% CaCO<sub>3</sub> is consistent with the dominance of eccentricity on deep ocean temperature (Figure 3b). Higher temperature at eccentricity maxima increases  $\Omega_{calcite}$  and lowers CaCO<sub>3</sub> solubility, promoting CaCO<sub>3</sub> preservation. Precession and obliquity cycles are consistent with the influence of ocean circulation and solubility-driven changes in CO<sub>2</sub> uptake on ocean chemistry. Changes in benthic ocean (CO<sub>3</sub><sup>2-</sup>), which also drive changes in  $\Omega_{calcite}$ , are dominated by precession and obliquity (Figure S9a).

The relative loss of high frequency power in wt% CaCO<sub>3</sub> compared to benthic ocean (CO<sub>3</sub><sup>2-</sup>) is also a function of calculating wt% CaCO<sub>3</sub> from the sedimentary mixed layer, that is, the upper ca. 5 cm of the sediment column in which mixing homogenizes the solid composition (Ridgwell, 2007a, 2007b). This mixing acts as a low-pass filter, amplifying eccentricity-driven cycles in wt% CaCO<sub>3</sub> arising from temperature-dependency relative to the precession and obliquity cycles from (CO<sub>3</sub><sup>2-</sup>)-dependency. Averaging sedimentary content over the upper 5-cm also induces a lag between 100 and 400 kyr eccentricity forcing and the global wt% CaCO<sub>3</sub> of 8 kyr (29°) and 9 kyr (7°), respectively.

Including solubility-related carbon feedbacks in the SYMM Exp.1 simulation results in similar power spectra compared to ASYM, despite the relative symmetry of ocean overturning (Figures 1c and 1d). Ocean overturning in SYMM is also primarily paced by precession (Figure S3c), creating distinct ~20 kyr cycles in benthic DIC  $\delta^{13}$ C (Figure S3e). However, precession power is absent in global mean wt% CaCO<sub>3</sub> (Figure S3f). Astronomically forced changes in ocean chemistry largely cancel out on precession timescales because the NH and SH overturning operate with roughly comparable strengths and in consequence, benthic ocean (CO<sub>3</sub><sup>2-</sup>) is dominated by eccentricity and absent of precession power (Figure S9b). The lag between 100 and 400 kyr eccentricity forcing and global CaCO<sub>3</sub> wt% in SYMM is 7 kyr.

#### 3.3. Marine Surface Productivity

Exp.2 elucidates how feedbacks involving marine productivity modifies the Earth system response to astronomical forcing. Global mean POC export production in ASYM (Figure 4b) is dominated by precession, illustrating the influence of changes in ocean circulation on surface nutrient supply and consistent with the seasonality of production. POC export also shows eccentricity modulation because elevated global ocean temperatures at eccentricity maxima result in stronger ocean overturning (Figure 3c), and elevated surface nutrient availability. CaCO<sub>3</sub> export shows the same power spectrum as POC export because of the use of a fixed rain ratio (Figure 4c).

Relative to Exp.1, precession cycles in  $pCO_2$  become more pronounced at the expense of eccentricity cycles (Figure 4d) which reduces the total  $pCO_2$  variability to ~26 ppm. Increased precessional power is a direct consequence of the impact that production and export of organic matter and carbonate has on surface DIC and ALK. The net impact is a decrease in mean annual surface ocean  $CO_2[aq]$  of 0.7% (or 0.173 µmol/kg) corresponding to maximum surface productivity. This translates roughly to a 0.7% reduction in  $pCO_2$ , equating to precession cycles with an average range of ~6 ppm, double the magnitude from Exp.1.

Changes in the magnitude and spatial pattern of export production do not notably change the power spectrum of benthic DIC  $\delta^{13}$ C compared to Exp.1 except for a slight increase in precession power (Figure 4e). The similarity in the carbon isotope response of Exp.1 and Exp.2 is because Exp.1 already incorporated the astronomically driven changes in isotope fractionation associated with export production as described in Section 3.1.

Sedimentary wt% CaCO<sub>3</sub> shows a slight loss of higher frequency power along with an increase in the maximum variability (Figure 4f). This shift in the power spectrum contrasts with the response of atmospheric  $pCO_2$  and benthic  $\delta^{13}C$  to the addition of varying export production. As CaCO<sub>3</sub> export increases in response to seasonal precession and obliquity forcing, surface and deep ocean  $[CO_3^{2-}]$  decreases, lowing  $\Omega_{calcite}$  and promoting CaCO<sub>3</sub> dissolution. Hence, even though CaCO<sub>3</sub> export and rain to the seafloor increases, the



**Figure 4.** Exp.2, marine surface productivity feedback in asymmetric configuration (ASYM). (a) Astronomical forcing parameters (as per Figure 3a). (b) Change in the export of particulate organic carbon. (POC). (c) Change in the export of  $CaCO_3$ . (d) Change in atmospheric  $CO_2$ . (e) Change in benthic  $\delta^{13}C$  of dissolved inorganic carbon (DIC). (f) Change in wt%  $CaCO_3$ . All variables are annual global mean values and are accompanied by their respective Fast Fourier Transforms (FFTs), normalized to the peak with maximum power. In blue, the values are plotted as anomalies with the previous experiment and depict the change driven by the marine productivity feedback only. Likewise, the FFTs are plotted as anomalies. A positive (green) value on the *y*-axis indicates an increase in relative power and a negative (red) value indicates a decrease in relative power compared to the previous experiment.



# Paleoceanography and Paleoclimatology



**Figure 5.** Exp.3, deep marine CaCO<sub>3</sub> feedback in asymmetric configuration (ASYM). (a) Astronomical forcing parameters (as per Figure 3a). (b) Change in benthic ocean temperature. (c) Change in the export of CaCO<sub>3</sub>. (d) Change in atmospheric CO<sub>2</sub>. (e) Change in benthic  $\delta^{13}$ C of dissolved inorganic carbon (DIC). (f) Change in wt% CaCO<sub>3</sub>. All variables are annual global mean values and are accompanied by their respective Fast Fourier Transforms (FFTs), normalized to the peak with maximum power. In blue, the values are plotted as anomalies with the previous experiment and depict the change driven by the marine CaCO<sub>3</sub> feedback only. Likewise, the FFTs are plotted as anomalies. A positive (green) value on the *y*-axis indicates an increase in relative power and a negative (red) value indicates a decrease in relative power compared to the previous experiment.

lower  $[CO_3^{2-}]$  prevents further CaCO<sub>3</sub> preservation. On eccentricity timecales, CaCO<sub>3</sub> export and benthic  $[CO_3^{2-}]$  vary in-phase and amplify short and long eccentricity power. The relative phasing between eccentricity forcing and CaCO<sub>3</sub> is unaffected by the responsive export productivity.

Global mean export production in SYMM Exp.2 (Figure S4c) varies less than ASYM Exp.2 (Figure 4c). In ASYM, the impact of precession is amplified because most of the export productivity occurs in the SH. Hemispheric productivity in SYMM is roughly balanced, which cancels out precession-induced changes and leaves eccentricity to dominate power spectra of atmospheric  $pCO_2$ . Variations in annual global mean  $pCO_2$ , benthic DIC  $\delta^{13}C$ , and CaCO<sub>3</sub> burial are muted, as a result of this hemispheric cancelation and there is little change with respect to Exp.1 (Figure S4d–S4f).

#### 3.4. Carbonate Compensation

We evaluate the impact of the carbonate compensation feedback in modifying the response to astronomical forcing in Exp.3 by now allowing burial of  $CaCO_3$  to modify the global ocean ALK inventory while no longer forcing weathering fluxes to always track burial, and contrast these results with Exp.2.

The total variability in atmospheric  $pCO_2$  increases to 64 ppm in Exp.3 and is relatively elevated during eccentricity maxima compared to Exp.2. The burial-weathering imbalance reduces the ocean carbon uptake capacity and this feedback impacts atmospheric  $pCO_2$  more than enhanced productivity. Thus,  $pCO_2$  shows enhanced eccentricity power and reduced power at high frequencies relative to Exp.2 (Figure 5d). Eccentricity cycles are significantly increased compared to Exp.2, with an average cycle range of almost 20 ppm for long eccentricity.

Consistent with  $pCO_2$ , power in benthic  $\delta^{13}C$  becomes slightly amplified at eccentricity frequencies in Exp.3 relative to Exp.2. The average range of long eccentricity cycles increases from 0.006‰ to ~0.011‰. Yet, obliquity and precession still remain the dominant cycles and control the total variability in DIC  $\delta^{13}C$  of 0.11‰.



# Paleoceanography and Paleoclimatology



**Figure 6.** Exp.4, terrestrial rock weathering feedback in asymmetric configuration (ASYM). (a) Astronomical forcing parameters (as per Figure 3a). (b) Change in surface land temperature. (c) Change in the rate of terrestrial weathering. (d) Change in atmospheric  $CO_2$ . (e) Change in benthic  $\delta^{13}C$  of dissolved inorganic carbon (DIC). (f) Change in wt% CaCO<sub>3</sub>. All variables are annual global mean values and are accompanied by their respective Fast Fourier Transforms (FFTs), normalized to the peak with maximum power. In blue, the values are plotted as anomalies with the previous experiment and depict the change driven by the terrestrial weathering feedback only. Likewise, the FFTs are plotted as anomalies. A positive (green) value on the *y*-axis indicates an increase in relative power and a negative (red) value indicates a decrease in relative power compared to the previous experiment.

Compared to Exp.0–2, increased CaCO<sub>3</sub> burial now removes ALK that is not resupplied by weathering (which is fixed), leading to lower deep ocean  $\Omega_{calcite}$  that counteracts increased burial. This results in reduced 100 and 400 kyr wt% CaCO<sub>3</sub> cycles in Exp.3 compared to Exp.2. The wt% reduction in 400 kyr cycles is much larger (from 0.7 to 0.2 wt%) than that of 100 kyr cycles (from 0.5 to 0.4 wt%) because the CaCO<sub>3</sub>-weathering imbalance is sustained over a much longer period that exceeds the residence time of carbon in the ocean, shifting the dominant spectral power from the 400 kyr to the 100 kyr eccentricity cycles (Figure 5f). The wt% CaCO<sub>3</sub> is now leading short eccentricity forcing by 5 kyr.

The change in the CaCO<sub>3</sub> preservation cycles in SYMM (Figure S5) are similar to ASYM because the impact of the ocean ALK imbalance on CaCO<sub>3</sub> at eccentricity maxima is similar. In SYMM too, this feedback amplifies 100 and 400 kyr cycles in  $pCO_2$  with the 400 kyr cycle producing the greatest increase in the average range. The long eccentricity remains the dominant  $pCO_2$  cycle with a value normalized to 1, and therefore does not show up as an increase in the  $\Delta$ FFT plot, contrary to ASYM in which dominant power shifted from precession to long eccentricity. The long eccentricity cycle in  $\delta^{13}$ C increases in average range from 0.002‰ to 0.003‰ and is considered negligible.

#### 3.5. Terrestrial Weathering Feedback

In our final experiment (Exp.4), we employ a temperature-dependent weathering parameterization (Colbourn et al., 2013) as opposed to keeping weathering rates invariant (Exp.3) or always balancing burial with weathering (Exp.0-2). As air temperatures over land rise, terrestrial silicate and carbonate weathering accelerate and thereby increase the transport of dissolved carbon and ALK to the surface ocean, allowing for greater carbon uptake and enhanced sedimentary  $CaCO_3$  preservation.

The power spectrum of annual global mean surface land temperature is dominated by 100 and 400 kyr eccentricity (Figure 6b). The relatively amplified precession power in land temperature compared to atmospheric temperature is a consequence of the asymmetric landmass distribution in ASYM. Land temperatures,

and therefore global weathering rates, are elevated when the NH is tilted maximally toward the Sun. Weathering rates consequently are dominated by precession cycles (Figure 6c).

Precession power in the weathering rate generates precession cycles in atmospheric  $pCO_2$  with an average range of 17 ppm, significantly larger compared to Exp.3. The average range of 400 kyr CO<sub>2</sub> cycles increases in comparison to Exp.3 to a value of 26 ppm. This relative increase in 400 kyr power is not apparent in the  $\Delta$ FFT plot (Figure 6d) because the power spectra are normalized, meaning that the dominant cycle (long eccentricity) maintains a value of 1 for Exp.3. The dominance of long eccentricity on atmospheric CO<sub>2</sub> is a result of the long ocean-atmosphere carbon residence time due to relatively slow weathering rates. The slow response time of the ocean's carbon reservoir acts as a low-pass filter (Cramer et al., 2003; Laurin et al., 2015; Pälike et al., 2006), further demonstrated by large phase differences of 112° and 125° between  $pCO_2$  and the short and long eccentricity forcing, respectively, that did not exist before. The total variability modeled in  $pCO_2$  increases to 132 ppm.

Maximum variability of 0.19‰ in benthic DIC  $\delta^{13}$ C arises from the imbalance between weathering  $\delta^{13}$ C and the  $\delta^{13}$ C of buried CaCO<sub>3</sub> (Figure 6e). Benthic DIC  $\delta^{13}$ C declines in response to elevated weathering and carbonate burial at long eccentricity maxima because relatively heavy carbon ( $\delta^{13}C_{CaCO3} = \delta^{13}C_{DIC} + 1.9\%$ ) is removed from the ocean through the burial of CaCO<sub>3</sub>. Burial of carbonate thus dominates the response of benthic DIC  $\delta^{13}$ C despite the input of high  $\delta^{13}$ C via weathering at eccentricity maxima. A high  $\delta^{13}$ C value for weathered carbonate is required in our model framework to balance volcanic carbon input because we do not simulate organic carbon burial. Inclusion of organic carbon burial would allow us to use a  $\delta^{13}$ C of weathered CaCO<sub>3</sub> more comparable to the  $\delta^{13}$ C of marine CaCO<sub>3</sub>. However, because the response of benthic DIC  $\delta^{13}$ C is negative at long eccentricity maxima, a lower value for the  $\delta^{13}$ C of weathered carbon would mean that weathering less effectively counteracts the impact on benthic DIC  $\delta^{13}$ C of increasing CaCO<sub>3</sub> burial and would drive an even greater decline in DIC  $\delta^{13}$ C.

Sedimentary wt%  $CaCO_3$  also shows a relative increase in power at the long eccentricity frequency and greater phase lag with eccentricity forcing, consistent with the slow weathering feedback. The wt%  $CaCO_3$  lags the short eccentricity forcing by 10 kyr.

The primary difference in SYMM simulations is the lack of precession power in the weathering rate and consequently in  $pCO_2$ , benthic DIC  $\delta^{13}C$ , and global mean wt% CaCO<sub>3</sub>. With landmasses distributed evenly, both the NH and SH have an equal amount of terrestrial area available for weathering, which minimizes the temporary imbalance on precessional timescales. Power spectra still show the relative shift toward long eccentricity forcing driven by the low pass filtering effect of the ocean carbon reservoir, consistent with ASYM (Figure S6).

#### 3.6. Astronomical Response to (Inorganic) Carbon Cycle Feedbacks

In addition to allowing us to elucidate the role of individual feedbacks, our experimental framework provides the net (Earth system) astronomical forcing response of ocean circulation,  $CO_2$  solubility, ocean productivity,  $CaCO_3$  dissolution and terrestrial weathering feedbacks combined (Figure 7 for ASYM and Figure S7 for SYMM). Under the modeled ice-free greenhouse climate state, ocean circulation,  $CO_2$  solubility, and  $CaCO_3$ -dissolution feedbacks are positive with respect to the astronomically driven temperature change and generate an increase in the average range in atmospheric  $CO_2$  of ~6 ppm (circulation and solubility) and ~14 ppm (19 minus 3.5 ppm) (CaCO<sub>3</sub> dissolution) at the long eccentricity period in ASYM. On the other hand, marine productivity reduces the average long eccentricity range in  $pCO_2$  by ~3 ppm (6.2 minus 3.5 ppm) and terrestrial weathering increases the range, but flips the phase, equating to a change of ~45 ppm (19 plus 26 ppm) in the long eccentricity cycle in ASYM (Table S2). Hence, marine productivity and terrestrial weathering feedbacks. The net  $\Delta pCO_2$  as simulated in Exp.4 is dominated by the terrestrial weathering feedback and slightly negative on eccentricity periods compared to a 0.30°C average range in Exp.0 that excludes all carbon-climate feedbacks. For SYMM, ocean circulation and solubility, and CaCO<sub>3</sub> dissolution increase the 400 kyr average range in  $pCO_2$  by 4.5 and 13 ppm, respective-





**Figure 7.** Cumulative astronomical impact of asymmetric configuration (ASYM) simulations. (a) Astronomical forcing parameters (as per Figure 3a). (b) Annual global mean temperature change in the benthic (black), and surface (blue) ocean. (c) Annual global mean  $pCO_2$  change. (d) Annual global mean  $\delta^{13}C$  change in the deep (black) and surface (blue) ocean dissolved inorganic carbon (DIC) reservoir. (e) Global mean sedimentary CaCO<sub>3</sub> change. All variables are accompanied by their respective Fast Fourier Transforms (FFTs), normalized to the peak with maximum power. The combined impact of all four climate-carbon feedbacks on atmospheric, ocean, and sedimentary reservoirs.

ly, whereas marine productivity decreases the range by 0.5 ppm and terrestrial weathering flips the phasing between eccentricity and  $pCO_2$ , equating to a change of 41 ppm (17 plus 24 ppm).

Our framework considers two groups of feedbacks, (a) astronomically forced changes in CO<sub>2</sub> solubility, ocean circulation, and marine surface productivity, which redistribute carbon between the ocean and atmosphere, and (b) astronomical forcing of carbonate compensation and silicate weathering feedbacks, which remove and add carbon to the atmosphere-ocean system. The feedbacks related to carbon redistribution act on timescales comparable to or smaller than those of global ocean overturning and therefore respond rapidly to astronomical forcing. Redistribution of carbon contributes to high frequency cyclicity produced in atmospheric  $pCO_2$ , global mean DIC  $\delta^{13}C$ , and global mean CaCO<sub>3</sub> preservation. However, the well-ventilated ocean simulated here, as a result of the simplified continental configurations likely results in a lower estimate of ocean circulation variability in response to astronomical forcing. For example, a more poorly ventilated ocean basin would store more carbon in the deep ocean reservoir and could release relatively more CO<sub>2</sub> to the atmosphere in response to small changes in ocean overturning strength, as shown with a theoretical framework along with a 3D ocean biogeochemistry model (Kwon et al., 2011) and more specifically for the Last Glacial Maximum using the cGENIE model (Ödalen et al., 2018).

By removing and adding carbon to the atmosphere-ocean system, carbonate compensation and silicate weathering feedbacks drive the most distinct responses in the carbon cycle, both in terms of the total variability in response to astronomical forcing and shifts in spectral power. Generally, long-term carbon feedbacks enhance power at low frequencies at the expense of high frequencies by acting as a low-pass filter at timescales comparable to the residence time of carbon in the ocean-atmosphere. This occurs in both ASYM and SYMM simulations and is therefore independent of the continental configuration. The spectral signals

in ASYM and SYMM of modeled  $pCO_2$ , benthic DIC  $\delta^{13}C$ , and wt% CaCO<sub>3</sub> are thus similar in the sense that the long eccentricity cycle dominates. However, a notable difference is the absence of precession cycles in SYMM (Figure S7) and their presence in ASYM (Figure S7) due to the hemispherically asymmetric impact of ocean circulation, solubility, and marine productivity on the redistribution of carbon between the surface and deep ocean in ASYM. High frequency cycles are particularly exacerbated in DIC  $\delta^{13}C$  of ASYM where marine productivity is greatest in the SH and local, seasonal changes in temperature and [CO<sub>2</sub> (aq)] affect the carbon isotope fractionation of POC and CaCO<sub>3</sub>.

While these simulations provide a systematic analysis of mechanisms through which astronomical forcing impacts the climate-carbon system, our experiments do not include a comprehensive representation of all relevant mechanisms. Our focus in this current paper is on the inorganic (carbonate) carbon cycle and feed-backs. We do not include a terrestrial biosphere (land plants and soils) or other reduced carbon reservoir (e.g., permafrost, methane hydrates). We also deliberately use the most simplistic parameterization of export productivity, so it is likely that we underestimate its sensitivity to astronomical forcing. Export productivity in these experiments is dependent only on the nutrient (phosphate) concentration with a modifier for light availability. Temperature dependence in parameterizations of export production of particulate organic matter and carbonate (Monteiro et al., 2012), and water column remineralization (Crichton et al., 2020; John et al., 2014) likely modify the carbon cycle response to astronomical forcing (see Crichton et al., 2020) for an analysis of the role of temperature in remineralization).

Moreover, we have applied a reflective boundary for organic matter and phosphate at the sediment-water interface, neglecting organic matter burial and assume an invariant nutrient inventory. Organic matter preservation and burial processes, in addition to reflecting changes in overlying export production, are temperature and oxygen dependent (Hülse et al., 2017), implying several different pathways by which astronomical forcing may additionally impact atmospheric  $pCO_2$  and DIC  $\delta^{13}C$ . Furthermore, inclusion of an open system for phosphate may result in distinct astronomical fingerprints as the phosphate residence time is similar in duration to high frequency astronomical cycles (van Cappellen & Ingall, 1996).

In comparison to benthic high-resolution paleoclimate records throughout the Cenozoic (Barnet et al., 2019; Beddow et al., 2016; Holbourn et al., 2013; Littler et al., 2014; Pälike et al., 2006; Tian et al., 2018; Westerhold et al., 2018), our simulations show minor variability in  $\delta^{13}$ C on astronomical timescales (the maximum variability in simulated  $\delta^{13}$ C cycles is less than 0.2‰ in ASYM). This model-data discrepancy is especially pronounced as we are evaluating modeled water column  $\delta^{13}$ C and not considering the impact of bioturbation on the range of preserved cycles. The muted simulated  $\delta^{13}$ C cycles very likely reflect missing feedbacks in the organic carbon cycle. Feedbacks in the marine organic carbon cycle will be addressed systematically in a follow-on paper.

Finally, an important caveat to our results is that, while cGENIE includes a dynamic ocean model, the 2D energy-moisture balance atmosphere does not reflect the potential impacts of astronomical forcing on atmospheric dynamics, for instance including changing patterns of winds or cloud cover. It is unclear exactly how inclusion of these dynamics in response to astronomical forcing would modify the changes in the global mean annual air temperature (MAAT) simulated here. However, the maximum variability in MAAT without carbon feedbacks in Exp.0 (1.7°C) corresponds well with the difference in MAAT between and eccentricity maxima and minima for the Eocene with  $2 \times (\sim 1.5^{\circ}C)$  and  $4 \times (\sim 2^{\circ}C)$  preindustrial CO<sub>2</sub> concentrations simulated with the HadCM3L GCM (Lunt et al., 2011).

#### 3.7. Implications for the Interpretation of Marine Proxy Records

While simulated cycles in benthic temperature,  $\delta^{13}$ C, and wt% CaCO<sub>3</sub> show modest ranges in comparison to proxy records of temperature (e.g.,  $\delta^{18}$ O), carbon isotopes ( $\delta^{13}$ C), and CaCO<sub>3</sub> from early Cenozoic ice-free climates, we can also compare model and data in terms of the relative distribution of power and phasing in key climate and carbon cycle metrics in response to astronomical forcing. Early Cenozoic deep-sea proxy records generally show pronounced 100 and 400 kyr eccentricity cycles (Barnet et al., 2019; Lauretano et al., 2015; Littler et al., 2014; Sexton et al., 2011; Westerhold et al., 2011; Zachos et al., 2010). Our results demonstrate that mean annual temperature spectra are dominated by eccentricity, despite the small impact





**Figure 8.** Phase analysis of paleoclimate proxies and comparable model output. (a and b) Benthic paleoenvironmental records from ODP Site 1262 of temperature ( $\delta^{18}$ O, in blue), stable carbon isotopes ( $\delta^{13}$ C, in green), and Fe counts (in orange). Below, the associated bandpass filtered signal of the long eccentricity cycle (320–480 kyr). (c and d) Simulated (negative) annual global mean deep ocean temperature (in blue),  $\delta^{13}$ C of dissolved inorganic carbon (DIC) (in green), and sedimentary carbonate weight percentage (in orange) of asymmetric configuration (ASYM) simulations Exp.4. Below, the associated bandpass filtered signal of the long eccentricity cycle.

of eccentricity on the insolation forcing. Strong 400 and 100 kyr power is also consistent with our modeled wt% CaCO<sub>3</sub> records, but 100 kyr power is notably lacking from modeled  $\delta^{13}$ C.

Importantly, modeled wt% CaCO<sub>3</sub> is the only model metric we report that is influenced by the low-pass filter effect of sedimentary mixing. Thus, CaCO<sub>3</sub> preservation is also dominated by 100 and 400 kyr cycles, despite the significance of precession in carbonate export and deep ocean chemistry. The spectral difference between carbonate export, deep ocean  $[CO_3^{2-}]$  and wt% CaCO<sub>3</sub> is hence also likely to shift power in proxy records of temperature and  $\delta^{13}$ C toward longer periods. However, this process cannot create power at the shorter 100 kyr period that is currently absent from modeled water column  $\delta^{13}$ C. Other processes, not included here, must be invoked to produce the 100 kyr  $\delta^{13}$ C eccentricity cycles in proxy records.

Early Cenozoic benthic foraminiferal records that contain evidence for astronomically paced "hyperthermals" (~60–48 Ma) also record characteristic phase relationships between temperature,  $\delta^{13}$ C, and CaCO<sub>3</sub>. Eccentricity-paced temperature maxima lead  $\delta^{13}$ C minima by 5–20 kyr and CaCO<sub>3</sub> minima often lead  $\delta^{13}$ C minima by ~2–40 kyr (Littler et al., 2014; Turner, 2014; Westerhold et al., 2018). However, our model simulations including all four feedbacks (Exp.4) show significant differences in the phase relationship between temperature, global mean wt% CaCO<sub>3</sub>, and the  $\delta^{13}$ C (Figure 8). Most strikingly, global mean wt% CaCO<sub>3</sub> is higher during eccentricity temperature maxima, a consequence in the model of elevated export production, warmer deep ocean temperatures, and elevated deep ocean [CO<sub>3</sub><sup>2–</sup>] caused by both stronger ocean overturning and enhanced terrestrial weathering. A few possibilities could explain this model-data discrepancy. First, the terrestrial weathering feedback might have been weaker than modeled here during the early Cenozoic (Caves et al., 2016; Kump & Arthur, 1997; van der Ploeg et al., 2018). Second, changes in shallow shelf carbonate burial might have compensated for deep sea burial. Changes in shelf burial might have corresponded to sea level fluctuations even in an ice-free world (Li et al., 2018; Liu et al., 2021). Third, additional carbon cycle feedbacks (not modeled) involving the organic carbon cycle might have dominated the phase relationship between temperature and carbonate preservation.

The hypothesis for a weaker weathering feedback has been postulated across climatic warming events including the Middle Eocene Climate Optimum (van der Ploeg et al., 2018) as well as the end-Permian (Kump, 2018). A gradual strengthening of the weathering feedback over the past 50 Myr has also been proposed to explain the long-term decline in  $pCO_2$  across the Cenozoic and to explain why the early Cenozoic was more sensitive to sudden carbon cycle perturbations (e.g., hyperthermals) than the mid and late Cenozoic (Caves et al., 2016). If silicate weathering rates are limited by factors like the thickness of soils (Kump, 2018), rock weatherability (Caves et al., 2016; van der Ploeg et al., 2018), or rate of rock uplift (Kump and Arthur, 1997) that prevent a strong positive response to increasing temperature, then the silicate weathering feedback would not have operated as modeled here in response to eccentricity-induced warming. Indeed, in Exp.3 with fixed terrestrial weathering, sedimentary carbonate content is relatively reduced during eccentricity-driven temperature maxima (compared to Exp.1, 2, and 4) when the weathering flux of carbonate ions to the ocean is insufficient to sustain CaCO<sub>3</sub> burial.

The second possibility is that the residual flux of the products of weathering to the open ocean were modulated by changes in shelf burial. An astronomical forcing that induces a greater shallow marine deposition response as compared to the total global weathering response, will result in the net supply to the open ocean becoming anti-phased. This could occur at least partly in response to changes in sea level (Li et al., 2018; Liu et al., 2021) and greater shelf area due to the thermal expansion of seawater (Lyle et al., 2008; Opdyke & Wilkinson, 1988), or directly from temperature controls on coral growth. However, the neritic system is complex and we lack a model representation of shallow water carbonate burial able to account for sea-level and accommodation space controls on neritic deposition.

Finally, the reduced carbon cycle may also play a key role in early Cenozoic carbon-climate variability, with, for instance, massive release of <sup>13</sup>C-depleted carbon into the ocean-atmosphere proposed to explain hyperthermal warming (e.g., Dickens, 2003; Lunt et al., 2011; DeConto et al., 2012; Komar, et al., 2013; Zachos et al., 2010; Zeebe et al., 2017). Given that our model framework currently neglects the dynamics of both terrestrial and marine reduced carbon reservoirs, we instead estimate the consequences of reduced carbon release in association with eccentricity maxima using previous cGENIE simulations without astronomical forcing. Vervoort et al., (2019) simulated 0.5% negative  $\delta^{13}$ C excursions with a duration of 100 kyr (comparable in size and duration to early Cenozoic  $\delta^{13}$ C cycles) via input of 600–1,500 Pg carbon with a  $\delta^{13}$ C signature of -22%. The required carbon input masses resulted from the prescribed carbon input durations between 25 and 75 kyr (or a maximum rate of 0.02 Pg yr<sup>-1</sup>). This forcing yields an increase in global temperature of  $0.5^{\circ}$ C–1°C. Yet, these "slow" perturbations to the carbon cycle do not generate notable changes in modeled wt% CaCO<sub>3</sub> since the temperature-dependent weathering feedback increases transport of al-kaline carbon species to the ocean at comparable rates (Vervoort et al., 2019). Carbon fluxes must exceed 0.2 Pg yr<sup>-1</sup> to offset enhanced weathering fluxes and drive dissolution coincident with warming.

Moreover, model-data discrepancy exists between the small 5–20 kyr lag between  $\delta^{18}$ O and  $\delta^{13}$ C in early Cenozoic proxy records, compared to the large lag of ~100 kyr simulated in cGENIE between benthic temperature and benthic DIC  $\delta^{13}$ C on eccentricity timescales (Figure 8a vs. 8d, Table S2). Large modeled lags between temperature and  $\delta^{13}$ C are produced only in experiments including CaCO<sub>3</sub> compensation and/or weathering feedbacks and occur because the small isotopic difference between inorganic carbon reservoirs requires long-sustained imbalances to produce a noticeable effect in the isotopic signature of the deep ocean DIC reservoir. Rapid input of low  $\delta^{13}$ C carbon would leave a notable imprint on the DIC reservoir much more rapidly due to the larger isotopic difference. Hence, geologically rapid release of low  $\delta^{13}$ C carbon at eccentricity paced intervals therefore fits best with the phase relationships observed between  $\delta^{18}$ O,  $\delta^{13}$ C, and CaCO<sub>3</sub> across astronomical timescales in early Cenozoic marine paleoclimate records and strongly points to the exclusion of dynamic reduced carbon reservoirs as the primary reason for model-data mismatch.



### 4. Conclusion

We have simulated the transient impact of astronomical forcing on the climate and (inorganic) carbon cycle using a dynamic 3D carbon-enabled Earth system model under ice-free conditions. Global annual mean temperature is dominated by eccentricity power with cycles of  $1.3^{\circ}C-1.7^{\circ}C$ , depending on the continental distribution. While eccentricity also dominates power spectra of early Cenozoic benthic foraminiferal  $\delta^{18}O$  records, observed variations correspond to cycles of  $2^{\circ}C-3^{\circ}C$  in deep ocean temperature, suggesting that additional feedbacks, not modeled here, roughly doubled the temperature response to insolation forcing.

We focused in this paper on feedbacks involving  $CO_2$  solubility, marine productivity,  $CaCO_3$  sedimentation, and terrestrial weathering. While these do not significantly amplify modeled temperature variability, they do cause distinct shifts in modeled carbon cycle proxies. The  $CO_2$  solubility feedback, ocean circulation, and marine primary production redistribute carbon between the atmosphere, surface, and deep ocean, producing high frequency (precession and obliquity) spectral signals in the benthic DIC  $\delta^{13}C$  reservoir. Long-sustained imbalances between  $CaCO_3$  sedimentation and terrestrial weathering dampen high frequency variability in  $\delta^{13}C$  and shift power spectra toward long eccentricity. Global mean wt%  $CaCO_3$  increases during eccentricity-driven warm intervals when enhanced weathering elevates ocean ALK. The opposite relationship between temperature and wt%  $CaCO_3$  in early Cenozoic records implies that the strength of the terrestrial weathering feedback was reduced, a shift in shallow shelf versus deep carbonate burial, and/or carbon cycle dynamics were controlled by additional (organic) carbon feedbacks. Geologically rapid, short-lived fluxes of reduced carbon into the atmosphere-ocean reservoir at eccentricity maxima are likely necessary to reconcile model-data discrepancies.

## **Data Availability Statement**

The version of the "muffin" release of the cGENIE Earth system model code used in this paper is tagged as release v0.9.25 and has is assigned a https://doi.org/10.5281/zenodo.5130666. Configuration files for the specific experiments presented here can be found in the directory: genie-userconfigs/MS/vervoortetal.2021. Details on the experiments, plus the command line needed to run each one, are given in the README.txt file in that directory. All other configuration files and boundary conditions are provided as part of the code release. A manual detailing code installation, basic model configuration, tutorials covering various aspects of model configuration and experimental design, plus results output and processing, is assigned a https://doi.org/10.5281/zenodo.5130672.

## References

- Barnet, J. S. K., Littler, K., Westerhold, T., Kroon, D., Leng, M. J., Bailey, I., et al. (2019). A high-fidelity benthic stable isotope record of Late Cretaceous–Early Eocene climate change and carbon-cycling. *Paleoceanography and Paleoclimatology*, 34. https://doi.org/10.1029/2019PA003556
- Beddow, H. M., Liebrand, D., Sluijs, A., Wade, B. S., & Lourens, L. J. (2016). Global change across the Oligocene-Miocene transition: High-resolution stable isotope records from IODP Site U1334 (equatorial Pacific Ocean). *Paleoceanography*, *31*(1), 81–97. https://doi. org/10.1002/2015pa002820
- Berger, A. (1978). Long term variations of daily insolation and quaternary climatic variations. *Journal of the Atmospheric Sciences*, 35(12), 2362–2367. https://doi.org/10.1175/1520-0469(1978)035<2362:ltvodi>2.0.co;2
- Bounceur, N., Crucifix, M., & Wilkinson, R. D. (2015). Global sensitivity analysis of the climate: Vegetation system to astronomical forcing : An emulator-based approach. *Earth System Dynamics*, 6, 205–224. https://doi.org/10.5194/esd-6-205-2015
- Brovkin, V., Bendtsen, J., Claussen, M., Ganopolski, A., Kubatzki, C., Petoukhov, V., & Andreev, A. (2002). Carbon cycle, vegetation, and climate dynamics in the Holocene: Experiments with the CLIMBER-2 model. *Global Biogeochemical Cycles*, *16*(4). https://doi. org/10.1029/2001GB001662
- Calov, R., Ganopolski, A., Petoukhov, V., Claussen, M., Brovkin, V., & Kubatzki, C. (2005). Transient simulation of the last glacial inception. Part II : Sensitivity and feedback analysis. *Climate Dynamics*, 24, 563–576. https://doi.org/10.1007/s00382-005-0008-5.40
- Caves, J. K., Jost, A. B., Lau, K. V., & Maher, K. (2016). Cenozoic carbon cycle imbalances and a variable weathering feedback. *Earth and Planetary Science Letters*, 450, 152–163. https://doi.org/10.1016/j.epsl.2016.06.035
- Colbourn, G., Ridgwell, A., & Lenton, T. M. (2013). The Rock Geochemical Model (RokGeM) v0.9. Geoscientific Model Development, 6(5), 1543–1573. https://doi.org/10.5194/gmd-6-1543-2013
- Cramer, B. S., Wright, J. D., Kent, D. V., & Aubry, M.-P. (2003). Orbital climate forcing of 8<sup>13</sup>C excursions in the late Paleocene-early Eocene (chrons C24n-C25n). Paleoceanography, 18(4), 1–25. https://doi.org/10.1029/2003PA000909
- Crichton, K. A., Wilson, J. D., Ridgwell, A., & Pearson, P. N. (2020). Calibration of key temperature-dependent ocean microbial processes in the cGENIE.muffin Earth system model (pp. 1–26). Geoscientific Model Development Discussions. https://doi.org/10.5194/gmd-2019-344
- DeConto, R. M., Galeotti, S., Pagani, M., Tracy, D., Schaefer, K., Zhang, T., et al. (2012). Past extreme warming events linked to massive carbon release from thawing permafrost. *Nature*, 484(7392), 87–91. https://doi.org/10.1038/nature10929
- Dickens, G. R. (2003). Rethinking the global carbon cycle with a large, dynamic and microbially mediated gas hydrate capacitor. *Earth and Planetary Science Letters*, 213(3–4), 169–183. https://doi.org/10.1016/s0012-821x(03)00325-x
- Edwards, N. R., & Marsh, R. (2005). Uncertainties due to transport-parameter sensitivity in an efficient 3-D ocean-climate model. *Climate Dynamics*, 24(4), 415–433. https://doi.org/10.1007/s00382-004-0508-8

#### Acknowledgments

While working on this manuscript P. Vervoort, S. Kirtland Turner, and A. Ridgwell were supported by a Heising-Simons Foundation award and NSF award 1702913. F. Rochholz was funded through the ERC Consolidator Grant "EarthSequencing" awarded to Prof. Dr. Heiko Pälike. A. Ridgwell also acknowledges support from the National Science Foundation under Grant 1658024.

- Ganopolski, A., Petoukhov, V., Rahmstorf, S., Brovkin, V., Claussen, M., Eliseev, A., & Kubatzki, C. (2001). Climber-2: A climate system model of intermediate complexity. Part II: Model sensitivity. *Climate Dynamics*, *17*(10), 735–751. https://doi.org/10.1007/s003820000144 Goosse, H., Brovkin, V., Fichefet, T., Haarsma, R., Huybrechts, P., Jongma, J., et al. (2010). Description of the Earth system model of inter-
- mediate complexity LOVECLIM version 1.2. Geoscientific Model Development, 3(2), 603–633. https://doi.org/10.5194/gmd-3-603-2010
  Gutjahr, M., Ridgwell, A., Sexton, P. F., Anagnostou, E., Pearson, P. N., Pälike, H., et al. (2017). Very large release of mostly volcanic carbon
  during the Palaeocene-Eocene Thermal Maximum. Nature, 548, 573–577. https://doi.org/10.1038/nature23646
- Heinemann, M., Timmermann, O., Elison, T., Saito, F., & Abe-Ouchi, A. (2014). Deglacial ice sheet meltdown: Orbital pacemaking and CO, effects. *Climate of the Past*, 10(4), 1567–1579. https://doi.org/10.5194/cp-10-1567-2014
- Holbourn, A., Kuhnt, W., Clemens, S., Prell, W., & Andersen, N. (2013). Middle to late Miocene stepwise climate cooling: Evidence from a high-resolution deep water isotope curve spanning 8 million years. *Paleoceanography*, 28(4), 688–699. https://doi. org/10.1002/2013pa002538
- Hülse, D., Arndt, S., Wilson, J. D., Munhoven, G., & Ridgwell, A. (2017). Understanding the causes and consequences of past marine carbon cycling variability through models. *Earth-Science Reviews*, 171, 349–382. https://doi.org/10.1016/j.earscirev.2017.06.004
- John, E. H., Wilson, J. D., Pearson, P. N., & Ridgwell, A. (2014). Temperature-dependent remineralization and carbon cycling in the warm Eocene oceans. Palaeogeography, Palaeoclimatology, Palaeoecology, 413, 158–166. https://doi.org/10.1016/j.palaeo.2014.05.019
- Komar, N., Zeebe, R. E., & Dickens, G. R. (2013). Understanding long-term carbon cycle trends: The late Paleocene through the early Eocene. *Paleoceanography*, *28*(4), 650–662. https://doi.org/10.1002/palo.20060
- Konijnendijk, T. Y. M., Weber, S. L., Tuenter, E., & van Weele, M. (2011). Past Methane variations on orbital timescales: A transient modeling experiment. Climate of the Past, 7(2), 635–648. https://doi.org/10.5194/cp-7-635-2011
- Kump, L. R. (2018). Prolonged Late Permian–Early Triassic hyperthermal: Failure of climate regulation? Philosophical Transactions of the Royal Society A: Mathematical, Physical & Engineering Sciences, 376(2130), 20170078. https://doi.org/10.1098/rsta.2017.0078
- Kump, L. R., & Arthur, M. A. (1997). Global chemical erosion during the Cenozoic: Weatherability balances the budgets. Tectonic uplift and climate change (pp. 399–426). Boston, MA: Springer. https://doi.org/10.1007/978-1-4615-5935-1\_18
- Kurtz, A. C., Kump, L. R., Arthur, M. A., Zachos, J. C., & Paytan, A. (2003). Early Cenozoic decoupling of the global carbon and sulfur cycles. Paleoceanography, 18(4). https://doi.org/10.1029/2003PA000908
- Kwon, E. Y., Sarmiento, J. L., Toggweiler, J. R., & Tim DeVries, T. (2011). The control of atmospheric pCO<sub>2</sub> by ocean ventilation change: The effect of the oceanic storage of biogenic carbon. *Global Biogeochemical Cycles*, *25*(3), 1–15. https://doi.org/10.1029/2011GB004059
- Laskar, J., Robutel, P., Joutel, F., Gastineau, M., Correia, A. C. M., & Levrard, B. (2004). A long-term numerical solution for the insolation quantities of the Earth. *Astronomy & Astrophysics*, *428*, 261–285. https://doi.org/10.1051/0004-6361:20041335
- Lauretano, V., Littler, K., Polling, M., Zachos, J., & Lourens, L. (2015). Frequency, magnitude and character of hyperthermal events at the onset of the Early Eocene Climatic Optimum. *Climate of the Past*, *11*(10), 1313–1324. https://doi.org/10.5194/cp-11-1313-2015
- Laurin, J., Meyers, S. R., Uličný, D., Jarvis, I., & Sageman, B. B. (2015). Axial obliquity control on the greenhouse carbon budget through middle- to high-latitude reservoirs. *Paleoceanography*, 30(2), 133–149. https://doi.org/10.1002/2014PA002736
- Li, M., Hinnov, L. A., Huang, C., & Ogg, J. G. (2018). Sedimentary noise and sea levels linked to land-ocean water exchange and obliquity forcing. Nature Communications, 9(1), 1–12. https://doi.org/10.1038/s41467-018-03454-y
- Littler, K., Zachos, J. C., Kersey, A., Roehl, U., & Westerhold, T. (2014). A high-resolution benthic stable-isotope record for the South Atlantic: Implications for orbital-scale changes in Late Paleocene-Early Eocene climate and carbon cycling. *Earth and Planetary Science Letters*, 401, 18–30. https://doi.org/10.1016/j.epsl.2014.05.054
- Liu, D., Huang, C., Kemp, D. B., Li, M., Ogg, J. G., Yu, M., & Foster, W. J. (2021). Paleoclimate and sea level response to orbital forcing in the Middle Triassic of the eastern Tethys. *Global and Planetary Change*, 199, 103454. https://doi.org/10.1016/j.gloplacha.2021.103454
- Liu, Z., Otto-Bliesner, B. L., He, F., Brady, E. C., Tomas, R., Clark, P. U., et al. (2009). Transient simulation of last deglaciation with a new mechanism for bolling-Allerod warming. Science, 325(5938), 310–314. https://doi.org/10.1126/science.1171041
- Lord, N. S., Ridgwell, A., Thorne, M. C., & Lunt, D. J. (2016). An impulse response function for the "long tail" of excess atmospheric CO<sub>2</sub> in an Earth system model. *Global Biogeochemical Cycles*, 30, 2–17. https://doi.org/10.1111/1462-2920.1328010.1002/2014gb005074
- Lourens, L. J., Sluijs, A., Kroon, D., Zachos, J. C., Thomas, E., Röhl, U., et al. (2005). Astronomical pacing of late Palaeocene to early Eocene global warming events. *Nature*, 435(7045), 1083–1087. https://doi.org/10.1038/nature03814
- Lunt, D. J., Ridgwell, A., luijs, A., Zachos, J., Hunter, S., & Haywood, A. (2011). A model for orbital pacing of methane hydrate destabilization during the Palaeogene. Nature Geoscience, 4, 775–778. https://doi.org/10.1038/ngeo1266
- Lyle, M., Barron, J., Bralower, T. J., Huber, M., Lyle, A. O., Ravelo, A. C., et al. (2008). Pacific Ocean and Cenozoic evolution of climate. *Reviews of Geophysics*, 46(2). https://doi.org/10.1029/2005rg000190
- Ma, W., Tian, J., Li, Q., & Wang, P. (2011). Simulation of long eccentricity (400-kyr) cycle in ocean carbon reservoir during Miocene Climate Optimum: Weathering and nutrient response to orbital change. *Geophysical Research Letters*, 38(10), 1–5. https://doi.org/10.1029/2011GL047680
- Maslin, M. A., & Ridgwell, A. J. (2005). Mid-Pleistocene revolution and the "eccentricity myth" (pp. 19–34). Geological Society Special Publication. https://doi.org/10.1144/GSL.SP.2005.247.01.02.43
- Menviel, L., Timmermann, A., Elison Timm, O., & Mouchet, A. (2011). Deconstructing the Last Glacial termination: The role of millennial and orbital-scale forcings. *Quaternary Science Reviews*, 30, 1155–1172. https://doi.org/10.1016/j.quascirev.2011.02.005
- Mix, A. C., Le, J., & Shackleton, N. J. (1995). 'Benthic foraminiferal stable isotope stratigraphy of Site 846: 0-1.8 Ma', Proceedings of the Ocean Drilling Program. Scientific Results, 138, 839–854.1991
- Monteiro, F. M., Pancost, R. D., Ridgwell, A., & Donnadieu, Y. (2012). Nutrients as the dominant control on the spread of anoxia and euxinia across the Cenomanian-Turonian oceanic anoxic event (OAE2): Model-data comparison. *Paleoceanography*, 27(4). https://doi.org/10.1029/2012pa002351
- Mook, W. G. (1986). <sup>13</sup>C in atmospheric CO<sub>2</sub>. Netherlands Journal of Sea Research, 20(2-3), 211-223. https://doi.org/10.1017/CBO9781107415324.00410.1016/0077-7579(86)90043-8
- Ödalen, M., Nycander, J., Oliver, K. I. C., Brodeau, L., & Ridgwell, A. (2018). The influence of the ocean circulation state on ocean carbon storage and CO<sub>2</sub> drawdown potential in an Earth system model. *Biogeosciences*, *15*(5), 1367–1393. https://doi.org/10.5194/bg-15-1367-2018
- Opdyke, B. N., & Wilkinson, B. H. (1988). Surface area control of shallow cratonic to deep marine carbonate accumulation. *Paleoceanog-raphy*, *3*(6), 685–703. https://doi.org/10.1029/pa003i006p00685
- Pälike, H., Norris, R. D., Herrle, J. O., Wilson, P. A., Coxall, H. K., Lear, C. H., et al. (2006). The heartbeat of the Oligocene climate system. Science, 314(5807), 1894–1898. https://doi.org/10.1126/science.1133822
- Peterson, C. D., Lisiecki, L. E., & Stern, J. V. (2014). Deglacial whole-ocean δ<sup>13</sup>C change estimated from 480 benthic foraminiferal records. *Paleoceanography*, 29, 549–563. https://doi.org/10.1002/2013PA002552

- Petoukhov, V., Ganopolski, A., Brovkin, V., Claussen, M., Eliseev, A., Kubatzki, C., & Rahmstorf, S. (2000). CLIMBER-2: A climate system model of intermediate complexity. Part I: Model description and performance for present climate. *Climate Dynamics*, *16*(1), 1–17. https://doi.org/10.1007/PL00007919
- Rau, G. H., Ulf, R., & Dieter, W.-G. (1996). A model of photosynthetic <sup>13</sup>C fractionation by marine phytoplankton based on diffusive molecular CO<sub>2</sub> uptake. *Marine Ecology Progress Series*, 133, 275–285. https://doi.org/10.3354/meps133275
- Ridgwell, A. (2007a). Application of sediment core modelling to interpreting the glacial-interglacial record of Southern Ocean silica cycling. Climate of the Past, 3(3), 387–396. https://doi.org/10.5194/cp-3-387-2007
- Ridgwell, A. (2007b). Interpreting transient carbonate compensation depth changes by marine sediment core modeling. *Paleoceanography*, 22(4), 1–10. https://doi.org/10.1029/2006PA001372
- Ridgwell, A., & Hargreaves, J. C. (2007). Regulation of atmospheric CO<sub>2</sub> by deep-sea sediments in an Earth system model. *Global Biogeo-chemical Cycles*, 21(2), 1–14. https://doi.org/10.1029/2006GB002764
- Ridgwell, A., Hargreaves, J. C., Edwards, N. R., Annan, J. D., Lenton, T. M., Marsh, R., et al. (2007). Marine geochemical data assimilation in an efficient Earth System Model of global biogeochemical cycling. *Biogeosciences*, 4(1), 87–104. https://doi.org/10.5194/bgd-3-1313-2006 Ridgwell, A., & Schmidt, D. N. (2010). Past constraints on the vulnerability of marine calcifiers to massive carbon dioxide release. *Nature*
- *Geoscience*, 3, 196–200. https://doi.org/10.1038/ngeo755 Ridgwell, A. J. (2001). *Glacial-interglacial perturbations in the global carbon cycle*. Diss: University of East Anglia.
- Sexton, P. F., Norris, R. D., Wilson, P. A., Pälike, H., Westerhold, T., Röhl, U., et al. (2011). Eocene global warming events driven by ventilation of oceanic dissolved organic carbon. *Nature*, 471(7338), 349–353. https://doi.org/10.1038/nature09826
- Short, D. A., Mengel, J. G., Crowley, T. J., Hyde, W. T., & North, G. R. (1991). Filtering of milankovitch cycles by earth's geography. Quaternary Research, 35(2), 157–173. https://doi.org/10.1016/0033-5894(91)90064-C
- Sigman, D. M., & Boyle, E. A. (2000). Glacial/Interglacial changes in atmospheric carbon dioxide. *Nature*, 407, 859–869. https://doi.org/10.1038/35038000
- Sloan, L. C., & Huber, M. (2001). Eocene oceanic responses to orbital forcing on precessional time scales. *Paleoceanography*, *16*(1), 101–111. https://doi.org/10.1029/1999PA000491
- Thompson, S. L., & Pollard, D. (1997). Greenland and Antarctic mass balances for present and doubled atmospheric CO<sub>2</sub> from the GENESIS version-2 global climate model. *Journal of Climate*, *10*, 871–900. https://doi.org/10.1175/1520-0442(1997)010<0871:GA AMBF>2.0.CO;2
- Tian, J., Ma, X., Zhou, J., Jiang, X., Lyle, M., Shackford, J., & Wilkens, R. (2018). 'Paleoceanography of the east equatorial Pacific over the past 16 Myr and Pacific–Atlantic comparison: High resolution benthic foraminiferal δ<sup>18</sup>O and δ<sup>13</sup>C records at IODP Site U1337. *Earth* and Planetary Science Letters, 499, 185–196. https://doi.org/10.1016/j.epsl.2018.07.025
- Turner, S. K. (2014). Pliocene switch in orbital-scale carbon cycle/climate dynamics. *Paleoceanography*, 29(12), 1256–1266. https://doi. org/10.1002/2014PA002651
- Turner, S. K., & Ridgwell, A. (2016). Development of a novel empirical framework for interpreting geological carbon isotope excursions, with implications for the rate of carbon injection across the PETM. *Earth and Planetary Science Letters*, 435, 1–13. https://doi.org/10.1016/j.epsl.2015.11.027
- Turner, S. K., Sexton, P. F., Charles, C. D., & Norris, R. D. (2014). Persistence of carbon release events through the peak of early Eocene global warmth. Nature Geoscience, 7(10), 748–751. https://doi.org/10.1038/NGEO2240
- van Cappellen, P., & Ingall, E. D. (1996). Redox stabilization of the atmosphere and oceans by phosphorus-limited marine productivity. *Science*, 271(5248), 493–496. https://doi.org/10.1126/science.271.5248.493
- van der Ploeg, R., Selby, D., Cramwinckel, M. J., Li, Y., Bohaty, S. M., Middelburg, J. J., et al. (2018). Middle Eocene greenhouse warming facilitated by diminished weathering feedback. *Nature Communications*, 9(1), 1–10. https://doi.org/10.1038/s41467-018-05104-9
- Vervoort, P., Adloff, M., Greene, S. E., & Kirtland Turner, S. (2019). Negative carbon isotope excursions: An interpretive framework. Environmental Research Letters, 14, 8085014. https://doi.org/10.1088/1748-9326/ab3318
- Wang, P., Tian, J., & Lourens, L. J. (2010). Obscuring of long eccentricity cyclicity in Pleistocene oceanic carbon isotope records. *Earth and Planetary Science Letters*, 290, 319–330. https://doi.org/10.1016/j.epsl.2009.12.028
- Weaver, A. J., Eby, M., Wiebe, E. C., Bitz, C. M., Duffy, P. B., Ewen, T. L., et al. (2001). The UVic Earth System Climate Model: Model description, climatology, and applications to past, present and future climates. *Atmosphere-Ocean*, 39(4), 361–428. https://doi.org/10.108 0/07055900.2001.9649686
- Westerhold, T., Röhl, U., Donner, B., McCarren, H. K., & Zachos, J. C. (2011). A complete high-resolution Paleocene benthic stable isotope record for the central Pacific (ODP Site 1209). *Global Biogeochemical Cycles*, 25(2), 1–13. https://doi.org/10.1029/2011WR010620
- Westerhold, T., Röhl, U., Donner, B., & Zachos, J. C. (2018). Global extent of early Eocene hyperthermal events: A new Pacific benthic foraminiferal isotope record from Shatsky Rise (ODP Site 1209). Paleoceanography and Paleoclimatology, 33(6), 626–642. https://doi. org/10.1029/2017PA003306
- Willeit, M., Ganopolski, A., & Feulner, G. (2013). On the effect of orbital forcing on mid-Pliocene climate, vegetation and ice sheets. Climate of the Past, 9(4), 1749–1759. https://doi.org/10.5194/cp-9-1749-2013
- Winkelmann, R., Levermann, A., Ridgwell, A., & Caldeira, K. (2015). Combustion of available fossil fuel resources sufficient to eliminate the Antarctic Ice Sheet. Science Advances, 1(8), 1–6. https://doi.org/10.1126/sciadv.1500589
- Xiao, D., Zhao, P., Wang, Y., & Zhou, X. (2013). Modeling the climatic implications and indicative senses of the Guliya [delta] <sup>18</sup>O-temperature proxy record to the ocean-atmosphere system during the past 130 ka. *Climate of the Past*, 9(2), 735. https://doi.org/10.5194/ cp-9-735-2013
- Zachos, J., Pagani, M., Sloan, L., Thomas, E., & Billups, K. (2001). Trends, rhythms, and aberrations in global climate 65 Ma to present. *Science*, 292(5517), 686–693. https://doi.org/10.1126/science.1059412
- Zachos, J. C., McCarren, H., Murphy, B., Röhl, U., & Thomas Westerhold, T. (2010). Tempo and scale of late Paleocene and early Eocene carbon isotope cycles: Implications for the origin of hyperthermals. *Earth and Planetary Science Letters*, 299, 242–249. https://doi. org/10.1016/j.epsl.2010.09.004
- Zeebe, R. E., Westerhold, T., Littler, K., & Zachos, J. C. (2017). Orbital forcing of the Paleocene and Eocene carbon cycle. *Paleoceanography*, 32(5), 440–465. https://doi.org/10.1002/2016PA003054
- Zeebe, R. E., & Wolf-Gladrow, D. (2001). CO<sub>2</sub> in seawater: Equilibrium, kinetics, isotopes (65th ed.). Gulf Professional Publishing
- Zhang, J., Quay, P. D., & Wilbur, D. O. (1995). Carbon isotope fractionation during gas-water exchange and dissolution of CO<sub>2</sub>. *Geochimica* et Cosmochimica Acta, 59(1), 107–114. https://doi.org/10.1016/0016-7037(95)91550-D